

# GOPEN ACCESS

**Citation:** Takahashi EA, Masoud L, Mukbel R, Guitian J, Stevens KB (2020) Modelling habitat suitability in Jordan for the cutaneous leishmaniasis vector (*Phlebotomus papatasi*) using multicriteria decision analysis. PLoS Negl Trop Dis 14(11): e0008852. https://doi.org/10.1371/journal. pntd.0008852

**Editor:** Guilherme L. Werneck, Universidade do Estado do Rio de Janeiro, BRAZIL

Received: May 22, 2020

Accepted: October 5, 2020

Published: November 23, 2020

**Copyright:** © 2020 Takahashi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript.

**Funding:** This work was part of the OIE Veterinary Education Twinning Programme between the Royal Veterinary College and the Jordan University of Science and Technology (grant number AD/VW/ 2017/365), funded by the UK Biosecurity Programme to ET and LM. The funders had no role in study design, data collection and analysis, RESEARCH ARTICLE

# Modelling habitat suitability in Jordan for the cutaneous leishmaniasis vector (*Phlebotomus papatasi*) using multicriteria decision analysis

Emi A. Takahashi<sup>1</sup>\*, Lina Masoud<sup>2</sup>, Rami Mukbel<sup>6</sup><sup>2</sup>, Javier Guitian<sup>1</sup>, Kim B. Stevens<sup>1</sup>\*

1 Royal Veterinary College, Hawkshead Lane, Hatfield, Hertfordshire, United Kingdom, 2 Department of Basic Medical Veterinary Science, Faculty of Veterinary Medicine, Jordan University of Science and Technology, Irbid, Jordan

\* emi.a.takahashi@gmail.com (ET); kstevens@rvc.ac.uk (KS)

# Abstract

Cutaneous leishmaniasis (CL) is a zoonotic vector-borne neglected tropical disease transmitted by female Phlebotomine sand flies. It is distributed globally but a large proportion of cases (70-75%) are found in just ten countries. CL is endemic in Jordan yet there is a lack of robust entomological data and true reporting status is unknown. This study aimed to map habitat suitability of the main CL vector, *Phlebotomus papatasi*, in Jordan as a proxy for CL risk distribution to (i) identify areas potentially at risk of CL and (ii) estimate the human population at risk of CL. A literature review identified potential environmental determinants for P. papatasi occurrence including temperature, humidity, precipitation, vegetation, wind speed, presence of human households and presence of the fat sand rat. Each predictor variable was (a) mapped; (b) standardized to a common size, resolution and scale using fuzzy membership functions; (c) assigned a weight using the analytical hierarchy process (AHP); and (d) included within a multicriteria decision analysis (MCDA) model to produce monthly maps illustrating the predicted habitat suitability (between 0 and 1) for P. papatasi in Jordan. Suitability increased over the summer months and was generally highest in the north-western regions of the country and along the Jordan Valley, areas which largely coincided with highly populated parts of the country, including areas where Syrian refugee camps are located. Habitat suitability in Jordan for the main CL vector—P. papatasi—was heterogeneous over both space and time. Suitable areas for P. papatasi coincided with highly populated areas of Jordan which suggests that the targeted implementation of control and surveillance strategies in defined areas such as those with very high CL vector suitability (>0.9 suitability) would focus only on 3.42% of the country's total geographic area, whilst still including a substantial proportion of the population at risk: estimates range from 72% (European Commission's Global Human Settlement population grid) to 89% (Gridded Population of the World) depending on the human population density data used. Therefore, high impact public health interventions could be achieved within a reduced spatial target, thus maximizing the efficient use of resources.

decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Cutaneous leishmaniasis (CL) is a globally distributed zoonotic vector-borne neglected tropical disease transmitted by female sandflies. The aim of this study was to estimate habitat suitability of the CL-transmitting sandfly in Jordan using spatial data on variables that influence the sandflies' presence and survival including temperature, humidity, precipitation, vegetation, wind speed, presence of human households and presence of the fat sand rat. Once high-risk areas were identified, the number of people at risk of infection could be estimated. Sandfly habitat suitability increased over the summer months and was generally highest in the north-western parts of the country and along the Jordan Valley, coinciding with highly populated regions, including areas containing Syrian refugee camps. These results suggest the need for targeted surveillance and disease control efforts.

# Introduction

#### Cutaneous leishmaniasis background and epidemiology

Cutaneous leishmaniasis (CL) is one of the 17 neglected tropical diseases (NTDs) identified by the World Health Organization (WHO) during the 10<sup>th</sup> meeting of the Strategic and Technical Advisory Group of Neglected Tropical Diseases in 2017 [1]. Despite causing a large disease burden among infectious diseases–Alvar *et al.* [2] estimated 1.2 million new cases of CL globally per year—CL is rarely included in tropical disease priorities. This is possibly due to its complex epidemiology and ecology, lack of current incidence data, lack of readily available tools for case management, and challenging treatment [2]. Cases are focally concentrated with approximately 70–75% of the estimated global CL incidence occurring in just ten countries namely Afghanistan, Algeria, Brazil, Colombia, Costa Rica, Ethiopia, Iran, Peru, Sudan and Syria [2].

CL is caused by an obligate intracellular protozoan from the genus Leishmania, which is spread through the bite of infected female *Phlebotomine* sandflies and is maintained by mammalian hosts [3,4]. There are more than 800 named sandfly species distributed globally of which 78 are currently known to be vectors for Leishmania spp. [5]. CL can either be anthroponotic if the natural reservoir of the parasite is human, or zoonotic if there is a suitable animal reservoir, of which the main ones are canids (e.g. dogs, foxes, jackals and wolves) and rodents (e.g gerbils, jirds and fat sand rats) [6]. There are numerous geographically specific Leishmania-sandfly combinations able to transmit CL, and 98 countries from five continents have reported endemic transmission [2,4,7]. However, some countries are considered endemic despite not reporting any human cases, due to detection of the Leishmania parasite in animal reservoir populations. For example L. major has frequently been isolated from gerbils in Mongolia but no human cases have been reported [2]. The fat sand rat (Psammomys obesus) is considered to be the main animal reservoir for L. major [8-10], with one study in Tunisia reporting up to 70% of trapped P. obesus testing positive for L. major [11]. P. obesus has a close ecological relationship with the sandfly Phlebotomus papatasi since the rodent's burrow provides sandflies with a moist and protected environment that is crucial for survival and reproduction. They offer adult sandflies a stable blood meal source, and larvae with necessary organic debris on which to feed and as a result P. papatasi is frequently found nearby P. obesus burrows [12-15].

#### Cutaneous leishmaniasis in Jordan: distribution and prevalence

CL is endemic in Jordan with the first case reported in 1929 [3]. It is mainly caused by *L. major* and transmitted by the *P. papatasi* sandfly [10]. There have been some sporadic CL cases caused by *L. tropica* in the northern border region, but it is relatively uncommon compared to *L. major* [16–18]. Although CL outbreaks have been regularly reported in different parts of the country [18–23], areas close to the Jordan Valley and the Dead Sea are considered the main endemic foci. For example, regions such as Swaimeh presented hyperendemicity characterized by 100% positive skin tests in individuals over the age of five in 1992 [21]. Moreover, detection of CL animal reservoirs like *P. obesus* has taken place in areas close to the Jordan Valley and the Dead Sea [15,24,25], which coincided with locations where CL cases have historically been reported [18,19,21–23].

In Jordan, CL is a notifiable disease and treatment is free. There is no national control programme and surveillance consists of passive case detection with weekly reports by physicians from local health departments (governorate level) to the Directorate for Disease Control, Ministry of Health (MoH) [26]. Although cases of CL have been reported annually, fluctuations in numbers may potentially represent variability in reporting capacity as severe under-reporting was described by Mosleh *et al.* [26] who detected a 47-fold difference between reported cases and those identified through active case detection.

Over the past few years, Jordan has witnessed a significant influx of Syrian refugees. Largescale human displacement and migration from non-endemic areas to areas with active CL transmission, or *vice versa*, are of particular concern regarding the association between conflict-terror and CL outbreaks [27]. This can result in more first-time exposure to CL, or the introduction of CL into new areas through susceptible sandfly vectors. Large-scale cross-border migration from Syria has taken place towards neighbouring countries such as Turkey, Lebanon, Jordan and Iraq [28], and CL cases have been reported in both refugee and non-refugee populations in Lebanon and Turkey [29,30]. Refugee camps and temporary settlements provide ideal conditions for enhanced transmission due to overcrowding, inadequate sanitation, waste disposal and inadequate housing, all conditions which characterised past CL outbreaks in refugee camps in Lebanon and Afghanistan [29,31]. Syrian refugee camps in Jordan are therefore at risk of CL outbreaks and should be considered within the scope of CL vector habitat suitability.

# Mapping CL distribution in Jordan

In 2014 the WHO prepared the '*Framework for action on cutaneous leishmaniasis in the Eastern Mediterranean Region 2014–2018*'. It was developed in consultation with regional member states and CL experts, aiming to strengthen surveillance, case management, disease prevention, capacity-building and research to reduce the burden of CL in the region. Proposed countrylevel actions included, amongst others: (i) the implementation of geographical information systems (GIS) to map and identify risk factors and areas of transmission risk, and (ii) the assessment of sandfly ecology to develop low environmental impact vector prevention and control plans [32].

Risk maps are valuable tools in public health that assist with decision-making by providing visual cues with which to prioritize and target prevention and control measures. The development of risk maps for vector borne diseases (including CL) often relies on vector exposure data instead of data on cases of infection to map risk. A primary advantage of using vector data over data on cases of infection includes the use of higher resolution data that are not restricted by patient data protection and anonymity. In addition, it overcomes the limitations associated with under-reporting, and uses accurate disease exposure locations which cannot be captured

by epidemiological data as these generally include patient domicile or reporting healthcare centre as locations [33]. Vector exposure risk maps have been developed using different GIS-based statistical models to estimate vector presence or abundance within a specific geographical area. Spatial models can be based on the spatial dependence of vector data (interpolation), or on the links between vector data and underlying environmental or socioeconomic variables (extrapolation). Spatial model outputs construct risk maps on a continuous surface and thus include information in areas that would be missing from surveillance programmes and epidemiological data [33,34]. Finally, the inclusion of population and demographic data can facilitate the estimation of the population at risk of vector exposure, which can be used as a proxy for risk of disease [35].

Spatial models previously built to explore the distribution and abundance of CL vectors generally comprised data-driven statistical models including general and generalized linear models [36–38], cluster analysis [39], principal component analysis [40] or ecological niche modelling [41–46]. However, these methods all require data-hence data-driven-from which to quantify associations between species' distribution and predictors, and therefore cannot be used when robust species distribution data are lacking, as in this instance.

In situations where georeferenced data describing a species distribution in environmental space with reference to a set of explanatory variables are not available, methods such as multicriteria decision analysis (MCDA), that use knowledge of the species distribution as model inputs-hence knowledge-driven-provide a valid alternative approach for generating habitat suitability maps that can be used to define the estimated distribution of a vector species [34] and inform targeted surveillance and control measures.

MCDA applies knowledge-driven rules through the implementation of Saaty's [47] analytical hierarchy process (AHP) and predictor-based fuzzy memberships to identify areas suitable for the occurrence of a vector or disease [48]. AHP is a framework which allows multi-layered decisions to be made-the comparative credible weighting of interwoven predictors, or criteria, with respect to species' habitat suitability—by reducing complex, multi-criteria comparisons (temperature versus rainfall versus humidity) to multiple simple pairwise comparisons (temperature versus rainfall; rainfall versus humidity; humidity versus temperature) to estimate the overall contribution of each predictor to habitat suitability. Advantages of using MCDA over data-driven modelling methods include its participatory element enabling the incorporation of multiple stakeholder perspectives and qualitative information into an unambiguous and transparent decision-making framework [34]. Furthermore, by not requiring the collection of data, this method is inexpensive and not subject to some of the traditional limitations and biases related to data collection methods [34,35]. Despite the increasing use of spatial MCDA in public health over the past decade, its application to vector-borne diseases remains limited [35,49-53]. However, in Jordan, where few studies have investigate the distribution of the CL vector, P. papatasi [15,24,54], and where there is a lack of robust national entomological data to feed such investigations, a knowledge-driven approach such as MCDA is an obvious and valid alternative for creating accurate risk maps in the absence of species' distribution data [34,55,56].

Although MCDA has been used to model the distribution of CL in north-eastern Iran [57], to the authors' knowledge this is the first study to use MCDA in the production of habitat suitability maps for leishmaniasis vector exposure, which could potentially guide and target interventions to minimise the burden of CL in Jordan.

#### Study aims and objectives

The aim of this study was to map habitat suitability of the main CL vector *P. papatasi* in Jordan, as a proxy for potential CL distribution, to (i) identify areas of highest CL risk and (ii)

estimate the human population at risk of CL infection in regions of highest vector suitability. This would help address some of the actions outlined within the WHO's '*Framework for action on cutaneous leishmaniasis in the Eastern Mediterranean Region 2014–2018*' [32] namely, recommendations regarding identification of high risk areas and targeted surveillance to minimise the burden of CL in Jordan.

# **Materials & Methods**

#### **Ethical Statement**

This study received ethical approval by the Clinical Research and Ethical Review Board at the Royal Veterinary College (URN 2017 1673–3).

#### Study area

The Hashemite Kingdom of Jordan is in the Middle East, approximately 100 km from the south-eastern region of the Mediterranean Sea, and shares borders with Israel, the State of Palestine, the Syrian Arab Republic, Iraq and Saudi Arabia. The country covers approximately 98,200 km<sup>2</sup> of land with a population of approximately 9.5 million in 2016 (not including the refugee population), subject to a 3.2% annual population growth [58].

Administratively, Jordan is divided into twelve governorates (Fig 1) and can also be divided into four biogeographic zones—Mediterranean, Irano-Turania, Saharo-Arabian and Sudania —which all vary geologically and biologically [59]. Western Jordan, where the Jordan Valley is located, has a Mediterranean climate with hot dry summers and cool wet winters. It is the most fertile region, containing the Jordan River, and has thus witnessed the most intense agricultural development in the country. Conversely, between 75–80% of the country is classified as arid or semi-arid receiving less than 100 mm of rain annually [59]. The Mountain Heights Plateau extends across the entire length of the western part of the country, separating the Jordan Valley from the eastern desert plains and contains Jordan's main population centres (Amman, Zarqa, Irbid and Karak). This area receives the highest rainfall and is therefore the most richly vegetated. Finally, the Badia region comprises approximately 75% of the country and is an area of desert and desert steppe, part of which is known as the North Arab Desert. Climate fluctuates widely with daytime summer temperatures exceeding 40°C, while winter nights can be very cold, dry and windy. Rainfall is minimal, averaging less than 50 mm annually [59].

#### Multicriteria decision analysis

The MCDA method. The steps involved in spatial MCDA have been described in detail elsewhere [35,48]. Briefly, this method consists of the identification of georeferenced risk factors associated with a specific outcome using knowledge (existing or hypothetical) of their relationship with the specified outcome (in this study, the outcome was the predicted presence of *P. papatasi* in Jordan). Spatial risk factor data layers are initially processed before inclusion in the model. All layers are standardized to have equal pixel size and resolution, and varying scales of the different predictors are standardized using fuzzy membership functions to achieve a common continuous scale. Fuzzy functions measure the degree of data cell membership within a predictor layer through predefined control points based on the association (known or hypothetical) between the predictor variable and the outcome. These associations define the shape (J-shaped, sigmoidal or linear) and direction (increasing, decreasing, or symmetric) [48] of the hypothesized relationship between predictor and outcome. The resulting standardized





layers are scaled across the entire range of values and the use of fuzzy functions operates on the basis that membership or non-membership of a location is gradual or can be partial [53,60].

If the model is to include only two equally important risk factors these can be combined using the Boolean operators "AND" or "OR". On the other hand, if the model is to include two or more risk factors that differ in importance, each risk factor can be assigned a weight using pairwise comparisons. This will break down information whereby only two criteria are considered at a time, and ultimately result in stronger risk factors influencing the modelled outcome to a larger degree [61]. The final modelled outcome map is a result of a suitability estimation calculated for each spatial unit (i.e. raster cell) using the following equation where

S = suitability,  $w_i$  = weight of factor *i*, and  $x_i$  = criterion score of factor *i*:

$$S = \sum w_i x_i$$

The suitability of each spatial unit is expressed on a continuous scale with lower values showing low suitability and high values representing areas of higher suitability.

Identification of predictor variables associated with Phlebotomous papatasi distribution. A literature search in Pubmed using the term 'Phlebotomus papatasi' identified 718 titles. These were screened at title, abstract and full text level and only publications in English that described the distribution and ecological conditions under which *P. papatasi* occurred and thrived were considered. Screening was carried out by a single reviewer (the first author) based on their subjective perception regarding the relevance of publications relating to bioecological information on *P. papatasi* occurrence. Title and abstract-level screening rejected studies that did not provide any information on spatial ecological determinants for P. papatasi occurrence and distribution. Rejected publications focused mostly on molecular mechanisms of Leishmania spp. infection, anatomical and morphological descriptions of P. papatasi, and the use of insecticides for vector control. Publications accepted after abstract screening (n = 135) were used to identify potential predictor variables while those accepted after full text screening (n = 52) were used to help define predictor range values for the model (median, minimum, maximum). These 52 publications included laboratory-based studies or field-based studies carried out in countries within North Africa and the Middle East (i.e. similar to Jordan).

The 135 publications used to identify potential predictor variables identified only seven spatial factors found to be associated with the occurrence and survival of *P. papatasi*, and taken forward to the MCDA model including temperature, humidity, precipitation, vegetation, wind speed, presence of human households and presence of the fat sand rat (*Psammomys obesus*). <u>Table 1</u> lists these seven predictor variables and the relationship between each and the occurrence of *P. papatasi*.

**Data sourcing and manipulation.** Georeferenced maps were sourced for each predictor variable where possible, and a proxy sought if there was no available georeferenced map for the original variable (Table 2). All maps were standardized to a common coordinate reference system (World Geodetic System 1984 (WGS\_84)) and resampled to a resolution of 0.00833 by 0.00833 decimal degrees. The resampling technique used for the categorical human settlement data was the nearest neighbour algorithm while a bilinear algorithm was used for all continuous data (the remaining variables). Jordan's administrative areas were obtained from the Global Administrative Areas website [62], and the country level administrative area was used to clip Jordan from global maps or region tiles. ArcMap 10.4.1 [63] was used for all spatial data manipulations.

WorldClim and CliMond use historical climate data and mathematical modelling techniques to create georeferenced raster maps for global mean monthly temperature, mean annual precipitation, mean monthly relative humidity and mean monthly wind speed [64,65]. In order to assess the degree of validity of modelled and actual climate data, simple linear regression analysis was carried out in R [66] between the modelled climate data (WorldClim and CliMond) and meteorological station data to measure the level of agreement between the two. National weather station data were requested from the Jordan Meteorological Department, and values underlying the 20 (24 for precipitation) weather station coordinates were extracted from temperature, relative humidity, precipitation and wind speed raster data maps (Table 3).

| Variable                              | Variable name                                    | Relationship with occurrence of Phlebotomus papatasi   |
|---------------------------------------|--|--|
| Mean monthly<br>temperature (C°)      | Temperature                                      | Sandflies are ectotherms and thus, temperature strongly influences developmental rates, survival and longevity [82, 83].   |
| Fat sand rat<br>distribution          | Psammomys obesus                                 | The fat sand rat ( <i>Psammomys obesus</i> ) is an important reservoir host for <i>Leishmania major</i> . <i>P. obesus</i> colonies and their burrows provide a source of food and refuge from extreme temperatures and desiccation [9,13,14,97]   |
| Mean annual<br>precipitation (mm)     | Precipitation                                    | Rainfall is associated with <i>P. papatasi</i> occurrence as it facilitates plant growth, which provides plant-based sugars for adult sandflies as well as food and shelter for rodents that also optimize <i>P. papatasi</i> survival [46].   |
| Mean monthly relative<br>humidity (%) | Relative humidity                                | <i>P. papatasi</i> require humid environments, especially larvae<br>and pupae that lack protective mechanisms against water<br>loss and are extremely susceptible to desiccation. Therefore,<br><i>P. papatasi</i> seek refuge and lay eggs in cracks within rocks<br>and houses, and rodent burrows. Furthermore, both larval<br>and adult stages' survival increases with higher relative<br>humidity levels [82]. |
| Monthly vegetation                    | Normalized difference<br>vegetation index (NDVI) | Vegetation provides plant-based sugars for adult sandflies<br>as well as food and shelter for rodents that also influence <i>P.</i><br><i>papatasi</i> occurrence [46]. It additionally is an indicator or<br>soil moisture, which optimizes larval survival within<br>burrows.  |
| Wind speed (m/s)                      | Wind   | Wind restricts sandfly activity and they are found more<br>abundantly in areas sheltered from wind or when wind<br>velocity is reduced [12,79,98,99].  |
| Human settlement                      | Human settlement                                 | <i>P. papatasi</i> benefits from human settlement as it provides<br>food sources and shelter. Studies have found positive<br>relationships between <i>Phlebotomus papatasi</i> occurrence and<br>the expansion of human settlements [84,100]. Furthermore,<br><i>P. papatasi</i> is considered to be highly endophilic<br>[45,79,81,100]; they are readily found in human domestic<br>environments.                  |

**Table 1. List of model predictor variables.** A description of variables identified by previous studies and publication sources as influencing the distribution of *Phlebotomus papatasi* and their relationship with its occurrence.

https://doi.org/10.1371/journal.pntd.0008852.t001

*P. obesus* distribution: As there was no available georeferenced map for the distribution of *P. obesus* in Jordan a proxy was sought. *P. obesus* is a specialist feeder and its occurrence is determined by the distribution of succulent plant species such as those of the *Chenopodiaceae* family [67,68]. The distribution of *Anabasis articulata* was thought to be a good proxy for *P. obesus* occurrence as it is an important component of *P. obesus*' diet in the Middle East [68–70]. Furthermore, *A. articulata* also forms part of *P. papatasi*'s diet [12]. To generate this map, a separate MCDA for *A. articulata* suitability distribution was performed. Soil properties describing soil texture and chemical composition (soil pH, percentage sand, clay and silt), and normalised difference vegetation index (NDVI) were identified from the published scientific literature as potential predictor variables for *A. articulata* occurrence [71].

Soil profile control points were defined using data from a study that measured soil properties at succulent vegetation growth sites, including *A. articulata*, in Egypt between 2004 and 2007 [71]. Georeferenced data were sourced for soil profiles in Jordan from the International Soil Reference and Information Centre (ISRIC) including soil pH, percentage sand, percentage clay and percentage silt [72]. El-Ghani *et al.* ([71]) carried out soil sampling between 0–50 cm deep thus, the ISRIC dataset was manipulated to create a new variable showing average values between soil layers D1 (0–19 cm) and D2 (20–39 cm). This new variable was joined to the

| Variable                       | Georeferenced map source  | Data manipulations   |
|--------------------------------|---|--|
| Temperature (C°)               | WorldClim version 2 at a 30 arc second spatial resolution (approximately 1 km <sup>2</sup> ) [64]             | Clip Jordan country administrative area from global map; standardize resolution.   |
| Psammomys obesus               | No map available so proxy was generated for inclusion in model  | MCDA for suitability distribution of the fat sand rat's main food source <i>Anabasis articulata</i> .  |
| Precipitation (mm)             | WorldClim version 2 at a 30 arc second spatial resolution (approximately 1 km <sup>2</sup> ) [64]             | Clip Jordan country administrative area from global map; calculate annual precipitation by the total sum of all months; standardize resolution |
| Relative humidity<br>(9am) (%) | CliMond climate dataset at a 10 arc minute spatial resolution [65]  | Clip Jordan country administrative area from global map; standardize resolution  |
| NDVI                           | Integrated Climate Data Centre (ICDC), Hamburg University [101].  | Clip Jordan country administrative area from region tile; calculate average NDVI between 2000–2016 for each month                              |
| Wind (m/s)                     | WorldClim version 2 at a 30 arc second spatial resolution (approximately 1 km <sup>2</sup> ) [64]             | Clip Jordan country administrative area from global map; standardize resolution  |
| Human settlement               | European Commission's Global Human Settlement Layer<br>(GHSL) at a 1 km <sup>2</sup> spatial resolution [102] | Clip Jordan country administrative area from global map; standardize resolution  |

#### Table 2. Georeferenced data sources and manipulations for predictor variables.

https://doi.org/10.1371/journal.pntd.0008852.t002

original dataset and individual raster maps for each soil property (soil pH, percentage sand, percentage clay and percentage silt) were created using the updated dataset.

An NDVI map with average values between April and September (see Section 3.2 for reasoning behind this selection of months) was also included in the MCDA. All maps were standardised to a common scale by applying fuzzy membership functions (see Section 2.2.4): a symmetric sigmoidal function was applied to each soil property using minimum, median and maximum values identified from the literature, as control points, and a monotonically increasing sigmoidal function was applied for NDVI values between 0.2 and 0.3 (corresponds to shrubs and therefore thought to most likely to represent *A. articulate*).

**Creation of fuzzy sets.** All maps were transformed and standardized to a common scale of 0–1 using fuzzy membership functions built on fuzzy logic [48,73] which assumes that membership of elements is not binary but can be gradual and/or partial (Table 4). In this way, uncertainty regarding the association between predictor variables and outcome is modelled using a fuzzy membership function to describe the shape of the hypothesized relationship between individual predictors and the outcome.

**Predictor weights.** Weights were calculated using Saaty's AHP [47] for both *P. obesus* and *P. Papatasi.* After standardisation to a common scale (0-1) predictor weights were determined using the pairwise comparison matrix of the AHP [61] (Table 5) whereby (i) predictor weights are derived by taking the principal eigenvector of a square reciprocal matrix of pairwise comparisons between the criteria, and (ii) a consistency ratio (CR) is calculated [61]. Pairwise comparisons were made using a 9-point scale (Table 5) comparing the subjective relative importance of the variable in the first column to the rest of the predictor variables. If the rating for the *yX* cell was 5 it would mean that *y* is strongly more important than *X* in determining the suitability of the outcome (i.e. the occurrence of *P. papatasi*). The combined total of all

Table 3. Climate data validation variables. Description of climate variables, number of weather stations, and years for which data were acquired from the Jordan Meteorological Department.

| Variable                       | Years                                    |
|--------------------------------|--|
| Mean monthly temperature       | 2010, 2012, 2014, 2016                   |
| Annual precipitation           | 2005–2011                                |
| Mean monthly relative humidity | 2010, 2012, 2014, 2016                   |
| Mean monthly wind speed        | 2004, 2006, 2008, 2010, 2012, 2014, 2015 |

https://doi.org/10.1371/journal.pntd.0008852.t003

| Predictor<br>variable | Rationale  |
|-----------------------|--|
| Temperature           | Laboratory experiments that determined <i>Phlebotomus papatasi's</i> thermal lower and upper limits suggested them to be 10°C when cold paralysis appeared, and 39.5°C when all individuals died [82]. However, at temperatures below 15°C no larvae reached pupal stage [83]. Field data from Iran, Egypt and Libya reported sandfly abundance peaks at temperatures between 25°C and 30°C [12,103–105]. Therefore, a symmetrical sigmoidal relationship was applied using the control points 15°C, 25°C, 30°C, 39.5°C.   |
| Precipitation         | Studies showed that there was lower probability of <i>Phlebotomus papatasi</i> occurrence when annual precipitation was lower than 50 mm [38, 44]. Abundance was recorded to be higher in areas where precipitation ranged between 100 mm to 600 mm with a maximum recording of 947 mm [44,46]. Therefore, a symmetrical sigmoidal relationship was applied using the control points 50 mm, 100 mm, 600 mm, 947 mm.  |
| NDVI                  | NDVI values range from -1 to +1; clouds and snow are characterized by negative values while<br>soils, rock and vegetation are positive. Barren areas of rock, sand or soils exhibit very low values<br>(0.1 to 0.2), shrub and grassland represent moderate values (0.2 to 0.3) while temperate and<br>tropical rainforests have high values (0.6 to 0.8) [106]. Therefore, in this study a linear<br>monotonically increasing relationship between 0.2 and 0.6 was applied. Higher values would<br>not be expected in arid or semiarid regions like Jordan, with the higher end values most likely<br>corresponding to agricultural land. |
| Relative humidity     | Adult <i>Phlebotomus papatasi</i> are able to survive at all relative humidity values 0–100%; however, the higher the relative humidity the higher the longevity [82]. Therefore, a monotonically increasing linear relationship ranging between 0% and 100% was applied.  |
| Psammomys<br>obesus   | <i>Psammomys obesus</i> provides shelter and food sources for <i>Phlebotomus papatasi</i> , which is found around <i>Psammomys obesus</i> ' burrows [9,13,14,97]. <i>Anabasis articulata</i> occurrence was used as a proxy for <i>Psammomys obesus</i> ' distribution[67,68], and a monotonically increasing linear relationship was applied using the MCDA output for the suitability distribution of <i>Anabasis articulata</i> . Control points were set at 0.1 and 1; the upper limit corresponding to the maximum MCDA value.  |
| Wind                  | Studies found sandfly activity to be highest when no wind was present, reduced when velocity was above 2 m/s and ceased when above 4 m/s [12,107,108]. Therefore, a monotonically decreasing sigmoidal relationship was applied using the control points 2 m/s and 4 m/s   |
| Human<br>settlement   | GHSL data was categorical; urban centre (3), urban cluster (2), and rural area (1). The positive relationship between <i>Phlebotomus papatasi</i> occurrence and human settlements [84,100] led to applying a monotonically increasing linear relationship from rural areas (1) to urban centres (2).  |

**Table 4. Predictor variables' fuzzy membership functions.** Inclusion of the associated rationale used to convert the predictor variables into fuzzy sets for inclusion in the multicriteria decision analysis model.

https://doi.org/10.1371/journal.pntd.0008852.t004

weights summed to one. Saaty (61) proved that for a consistent reciprocal matrix, the largest Eigenvalue ( $\lambda_{max}$ ) needed to be equal to the size of the comparison matrix (n), or  $\lambda_{max} = n$ . Based on this principle, he went on to propose a method to calculate a consistency ratio that included a consistency index (CI) and random consistency index (RI). The CI is a measure of consistency and is defined as:

$$CI = \frac{\lambda_{max} - n}{n - 1}$$

The CI is then compared to the RI, which is a randomly generated reciprocal matrix that uses different scales in a similar way to bootstrapping. Finally, the consistency ratio (CR) is defined as

$$CR = \frac{CI}{RI}$$

The CR measures the consistency of judgements in relation to large samples of random judgements [61]. If CR>0.1 the judgements are close to randomness and the AHP should be re-evaluated.

| Pairwise comparison 9-point continuous rating scale |               |                  |            |         |            |          |               |           |
|---|---------------|------------------|------------|---------|------------|----------|---------------|-----------|
| 1/9   | 1/7           | 1/5              | 1/3        | 1       | 3          | 5        | 7             | 9         |
| Extremely   | Very strongly | Strongly         | Moderately | Equally | Moderately | Strongly | Very Strongly | Extremely |
| LESS IMPORTANT                                      |               |                  |            | MORE I  | MPORTANT   |          |               |           |
|   |               |                  |            |         |            |          |               |           |
|   | Pre           | dictor variables |            |         | X          |          | Y             | Z         |
| X   |               |                  |            |         |            |          |               |           |
| y   |               |                  |            | yХ      |            |          |               |           |
| Z   |               |                  |            | zX      |            | zY       |               |           |

#### Table 5. Rating scale used for pairwise comparisons between predictor variables.

https://doi.org/10.1371/journal.pntd.0008852.t005

Owing to the subjective nature of the pairwise comparison method, an expert elicitation was used initially to minimize bias. Experts within the field of *P. papatasi* ecology in the Middle East were identified from the previously conducted literature review to identify *P. papatasi* predictor variables, and through international conferences such as WorldLeish2017 (Toledo). Six experts were contacted via email and provided with a brief explanation of the objectives and methods of the study along with an uncompleted pairwise comparison table (Table 6). In addition to rating the relative importance between predictor variables in relation to the specified outcome using the scale presented in Table 5, a confidence score ranging between 1 (not confident) to 5 (most confident) was to be included by the experts for each rating within the matrix to quantify the assessor's confidence in their score. The final pairwise comparison matrix from which predictor weightings were calculated, used the weighted geometric mean of each of the expert's subjective opinion using the confidence score as a weight:

$$\bar{x} = \exp\left(\frac{\sum_{i=1}^{n} w_i \ln x_i}{\sum_{i=1}^{n} w_i}\right)$$

where  $x_i$  = pairwise comparison rating

 $w_i = \text{confidence score}$ 

However, as the expert elicitation resulted in a CR>1 for all experts and were therefore considered to be little better than random, it was decided that pairwise comparisons for both *P. obesus* and *P. Papatasi* should instead be made by the first author, who had completed the literature review of factors associated with vector occurrence and survival

**MCDA model.** Fuzzy sets and predictor weightings were obtained using the Terrset Geospatial Monitoring and Modelling software [74]. Once all six predictor variables were (a)

| Table 6. Pairwise comparison matrix of the analytical hierarchy process (AHP) for the predictors associated with the occurrence of Phlebotomus papatasi in Jord | lan |
|---|-----|
| Based on the first author's subjective judgment constructed from the literature review in Section 2.2.2.  |     |

|                   | Temp | Relative humidity | Precip | NDVI | Psammomys obesus | Wind | H. Sett. | Weight |
|-------------------|------|-------------------|--------|------|------------------|------|----------|--------|
| Temperature       | 1    |                   |        |      |                  |      |          | 0.3590 |
| Relative humidity | 1/7  | 1                 |        |      |                  |      |          | 0.0332 |
| Precipitation     | 1/5  | 3                 | 1      |      |                  |      |          | 0.0615 |
| NDVI              | 1/5  | 3                 | 1      | 1    |                  |      |          | 0.1210 |
| Psammomys obesus  | 1/3  | 5                 | 3      | 3    | 1                |      |          | 0.1210 |
| Wind              | 1/3  | 3                 | 3      | 3    | 3                | 1    |          | 0.1774 |
| Human settlement  | 1/3  | 5                 | 3      | 3    | 3                | 1    | 1        | 0.1864 |

\* Consistency ratio = 0.05; Temp = temperature; Precip = precipitation; H. Sett = human settlement; NDVI = normalized difference vegetation index

https://doi.org/10.1371/journal.pntd.0008852.t006

identified; (b) sourced and georeferenced; (c) standardized to a common size, resolution and scale using fuzzy membership functions; and (d) weighted using the AHP, the MCDA model was run to produce a final map illustrating the predicted habitat suitability for *P. papatasi* in Jordan. Suitability values ranged between 0 and 1 with higher values indicating higher suitability. Individual suitability maps were produced for the months between April and September (see Section 3.2 for the rationale behind selection of these months).

**Sensitivity analysis.** A sensitivity analysis was carried out to explore how changes in weights and membership functions for each predictor variable would affect the final suitability estimate. To investigate the effect of membership function, the model was run assuming a linear association between all predictor variables and the occurrence of *P. papatasi*. Likewise, to investigate the effect of a change in weights, the model was run with equal weights applied to all predictor variables. Final suitability estimates were extracted from 100 random points from the original and new maps, and mean changes between these values were compared using Student's t-tests (histograms and QQ plots showed all variables were normally distributed) using IBM SPSS Statistics 23.

# Human population at risk of P. papatasi exposure

Binary maps depicting high versus low *P. papatasi* suitability were created using different cutoff values (0.5, 0.6, 0.7, 0.8 and 0.9) for each month between April and September, before taking an average of the six months for each cut-off value. The population at risk (PAR) of *P. papatasi* exposure was defined as the total population residing in areas identified as suitable (above 0.5, 0.6, 0.7, 0.8 or 0.9) for *P. papatasi* occurrence. To calculate this, two different gridded population maps were used: the European Commission's Global Human Settlement (GHS) population grid (http://ghsl.jrc.ec.europa.eu/ghs\_pop.php), and the Gridded Population of the World (GPW) (http://sedac.ciesin.columbia.edu/data/set/gpw-v4-populationcount-adjusted-to-2015-unwpp-country-totals). Different population datasets were used to account for the uncertainty associated with different gridded human population datasets [75].

# Results

#### Psammomys obesus distribution

The AHP assigned a weight of 0.5556 for vegetation and 0.1111 for each of the soil properties (Table 7). The resulting MCDA produced the estimated distribution of *A. articulata* displayed in Fig 2. Suitability values ranged between 0–1 with lower values (green) indicating lower suitability, and conversely higher values (red) indicating higher suitability for occurrence of *A. articulata*. High suitability areas were concentrated in the north-west region of the country.

# Climate data validation

Linear regression between weather station data and modelled climate data gave different results for different variables. Mean monthly temperature and mean annual precipitation showed the highest levels of association and agreement (Fig 3A–3D) as demonstrated by their high R<sup>2</sup> values (Table 8). For mean monthly relative humidity and mean monthly wind speed, the association was statistically significant for most months (Fig 3B and 3C) but R<sup>2</sup> values fluctuated and were lower compared to mean monthly temperature and mean annual precipitation (Table 8). The results also showed that data on warmer months were more reliable, indicated by higher R<sup>2</sup> values (Table 8). The suitability model was therefore run only for the months between April and September. This coincided with the timeframe of most interest from an epidemiological perspective, as this is when *P. papatasi* is most abundant [76–78].

|          | NDVI  | pН | Clay (%) | Sand (%) | Silt (%) | Weight |
|----------|-------|----|----------|----------|----------|--------|
| NDVI     | 1     |    |          |          |          | 0.5556 |
| рН       | 1/5   | 1  |          |          |          | 0.1111 |
| Clay (%) | 1/5   | 1  | 1        |          |          | 0.1111 |
| Sand (%) | 1/5   | 1  | 1        | 1        |          | 0.1111 |
| Silt (%) | 1/5   | 1  | 1        | 1        | 1        | 0.1111 |
|          | · · · |    | ·        |          |          |        |

Table 7. Pairwise comparison matrix of the analytical hierarchy process (AHP) for the predictors associated with the occurrence of *Psammomys obesus* in Jordan. Based on the first author's subjective judgement constructed from published literature.

\* CR = 0.00

https://doi.org/10.1371/journal.pntd.0008852.t007

#### **Predictor variables**

Fuzzy membership functions were applied to each predictor variable. Moreover, for those predictor variables for which monthly data were available, fuzzy membership functions were applied to each month between April and September. Values ranged from 0 (green) to 1 (red) indicating the degree of membership of suitability of each raster cell for P. papatasi occurrence (Fig 4). Membership for *P. papatasi* occurrence suitability in terms of annual precipitation (Fig 4A), was widespread but low in the south and south-eastern parts of the country, where rainfall is scarce. For A. articulata (proxy for P. obesus) (Fig 4B) membership was higher in the north-western tip of the country. Finally, membership regarding human settlement (Fig 4C) was also concentrated in the north-western parts of the country in line with the location of the largest most populated cities such as Amman, Irbid and Zarqa. Monthly fluctuations concerning cell membership for *P. papatasi* suitability were observed for the remaining environmental predictor variables. Larger areas of the country became suitable in terms of temperature as summer progressed (Fig 4D). Cell membership associated with relative humidity was higher in the north-west but became more widespread throughout the country during July (Fig 4E) while cell membership linked to vegetation was focussed mostly in the north-western tip of the country and decreased over the summer months (Fig 4F). Cell membership associated with wind showed a peak along the Jordan Valley at the western border of the country with higher widespread membership in September due to reduced wind speed for this month (Fig 4G).

# Weights

The completed pairwise comparison matrix for the AHP is detailed in Table 6. The highest weighted predictor variable was temperature followed by human settlement and wind speed, while relative humidity was the lowest weighted predictor variable. The consistency ratio was 0.05.

# Suitability distribution of P. papatasi in Jordan

Suitability for the occurrence of *P. papatasi* in Jordan for the different months between April and September is presented in Fig 5 using a continuous 0 to 1 scale. Blue areas were areas of very low or low suitability; yellow and light orange areas were moderately suitable, while dark orange and red areas were considered highly suitable for *P. papatasi* occurrence. In general, suitability was higher in the north-western areas and along the Jordan Valley to the west, while suitability was generally lower in the eastern and southern regions of the country. Suitability increased with the progression of summer and September showed the highest and most widespread suitability. Syrian refugee camps (depicted as black stars) were in areas of low to





moderate *P. papatasi* suitability except for September. However, those located more towards the northwest were in very close proximity to areas displaying high suitability.

The extent of geographic areas suitable for *P. papatasi* occurrence varied according to suitability cut-offs; areas defined by a higher suitability cut-off were smaller and more dispersed (Fig 6A and 6B) compare to areas defined by a lower or moderate suitability cut-off (Fig 6C–6E), which were more homogenous and extensive.





#### Sensitivity analysis

In general, the change in fuzzy membership functions caused a larger mean change than changing the weights (Table 9). Equal weights of 0.143 for each of the seven predictor variables produced an overall mean change of  $0.06 \pm 0.13$  units, while applying a linear relationship for all predictor variables resulted in a mean change of  $0.09 \pm 0.16$  units.

#### Population at risk (PAR)

The use of the two population grids (GHS and GPW) to calculate PAR for different months and at different suitability cut-offs, gave a similar pattern but at sizeably different scales (Fig 7). PAR was positively associated with the progression of summer; the number of people at risk of *P. papatasi* exposure was lowest in April and increased throughout the summer months with similar values during June, July and August, and finally September when PAR was highest. The GPW population grid presented a total national population of 7,536,579, while the GHS population grid presented a total national population of 4,600,878; a 39% difference. This translated into the GPW-based PAR calculation being of a substantially larger order of magnitude compared to GHS (millions versus hundreds of thousands; Fig 7).

|            | Manna and have been been bits  | M                       | Maria mandila dama andara | M                         |
|------------|--------------------------------|-------------------------|---------------------------|---------------------------|
| Month/Year | Mean monthly relative numidity | Mean monthly wind speed | Mean monthly temperature  | Mean annual precipitation |
| January    | 0.31*                          | 0.02°                   | 0.7*                      |                           |
| February   | 0.3*                           | 0.15*                   | 0.73*                     |                           |
| March      | 0.4*                           | 0.14*                   | 0.71*                     |                           |
| April      | 0.35*                          | 0.3*                    | 0.86*                     |                           |
| May        | 0.4*                           | 0.3*                    | 0.9*                      |                           |
| June       | 0.38*                          | 0.24*                   | 0.84*                     |                           |
| July       | 0.42*                          | 0.22*                   | 0.88*                     |                           |
| August     | 0.38*                          | 0.28*                   | 0.85*                     |                           |
| September  | 0.27*                          | 0.31*                   | 0.86*                     |                           |
| October    | 0.13*                          | 0.31*                   | 0.79*                     |                           |
| November   | 0.02                           | 0.1*                    | 0.67*                     |                           |
| December   | 0.17*                          | 0.01                    | 0.72*                     |                           |
| 2005       |                                |                         |                           | 0.83*                     |
| 2006       |                                |                         |                           | 0.88*                     |
| 2007       |                                |                         |                           | 0.89*                     |
| 2008       |                                |                         |                           | 0.83*                     |
| 2009       |                                |                         |                           | 0.87*                     |
| 2010       |                                |                         |                           | 0.72*                     |

**Table 8.** Climate data validation results. R<sup>2</sup> values derived from the simple linear regression analysis between weather station climate data recordings and modelled climate data using historical recordings. Overall, R<sup>2</sup> values were higher for the warmer months of the year (April-September).

\* P-value < 0.001

° P-value < 0.05

https://doi.org/10.1371/journal.pntd.0008852.t008

However, the change in PAR in response to different suitability cut-offs was not proportional to the change observed for geographic range at risk. There was a large reduction in geographical area at risk ranging from 57% of the country at moderate (0.5) suitability, to 3% of the country at very high suitability (0.9). However, the change in PAR was more modest; according to GHS and GWP population grids 13% and 100% of the total Jordanian population were at moderate (0.5) risk, while 10% and 90% were at very high risk (0.9) respectively (Table 10).

#### Discussion

This is the first study, to the author's knowledge, to explore the distribution of the main CL vector *P. papatasi* in Jordan. It adds to the limited number of studies that have investigated CL vector distribution in the Middle East using spatial analysis tools, none of which used MCDA [41,44–46]. In addition, this study separates itself from past studies investigating CL vectors by including a set of carefully selected variables using evidence-based biological rationale rather than being restricted to variables for which data are available. Other predictive studies on CL vector distribution have mostly included readily available variables such as the WorldClim bioclimatic variables [41–43,45,46], which are easily accessible and provide a range of resolutions, the highest being 1 km<sup>2</sup>. Despite the quality of WorldClim data, of the available 19 bioclimatic variables, 11 are temperature-related and the remaining 8 are precipitation-related. Therefore, the diversity of environmental variables included in previous predictive models is limited.

In addition to the inclusion of a comprehensive range of biologically-backed predictor variables, this study calculated the population at risk of CL, which has never been undertaken in Jordan. This is a key metric considering the current demographic changes that are taking place in Jordan, especially with conflict-related displacement in neighbouring countries.



**Fig 4. Fuzzy membership maps for each predictor variable.** (A) Annual precipitation, (B) *Anabasis articulata*, (C) human settlement, (D) temperature, (E) relative humidity, (F) vegetation and (G) wind.

https://doi.org/10.1371/journal.pntd.0008852.g004

#### Habitat suitability for P. papatasi occurrence

The results showed higher habitat suitability for CL vector occurrence in the north-western areas and along the Jordan Valley to the west of the country. *P. papatasi* have previously been found in neighbouring adjacent regions along the Dead Sea and the Jordan Valley in Israel and the State of Palestine [9,14,79–81], which reaffirms the suitability of these nearby areas to sustain *P. papatasi*. Additionally, regions most suitable for sustaining *P. papatasi* showed broad agreement with locations where CL cases have been reported historically [18,19,21–23].

The north-west region was suitable for *P. papatasi* occurrence as it presented optimal predictor variable values as shown in the fuzzy membership function maps (Fig 4). North-western areas had favourable temperatures, higher relative humidity, suitable locations for *P. obesus*, higher vegetation index, increased human settlement, higher annual precipitation, and lower wind speed. *P. papatasi's* predicted distribution range increased throughout the summer months, which was expected due to the warmer temperatures enabling and optimising sandfly development [82,83]. The largest geographical range of high suitability areas was found in September due to the lower recorded wind speed, which led to more widespread favourable conditions.

High *P. papatasi* suitability areas coincided with the most populated regions in Jordan. This is of concern due to *P. papatasi*'s endophilic nature, being found indoors within domestic environments more frequently than other sandfly species [79,81,84]. This behavioural trait, in



Fig 5. MCDA outputs for the predicted suitability distribution of *Phlebotomus papatasi* in Jordan from April to September. \* show the location of Syrian refugee camps.

https://doi.org/10.1371/journal.pntd.0008852.g005



Fig 6. Geographic areas suitable for Phlebotomus papatasi occurrence using suitability cut-offs of 0.9 to 0.5.

addition to facilitating CL transmission, could benefit *P. papatasi* survival by providing stable optimal conditions with minimal competition, as other sandfly species are not as endophilic. On the other hand, it could be advantageous from a vector control perspective as indoor residual spraying could be largely effective against an endophilic vector.

#### Use of risk maps in cutaneous leishmaniasis mitigation strategies in Jordan

CL is an endemic and notifiable disease in Jordan. However, there is no national leishmaniasis control program and case detection is passive. The results from this study are valuable for addressing the actions outlined within the WHO East Mediterranean Region's framework. Firstly, risk factors were synthesized, weighted and mapped. Furthermore, the integration of *P. papatasi* risk factors into a knowledge-driven habitat suitability model, identified risk areas for CL vector occurrence and thereby, CL transmission risk. The results revealed high human population densities in *P. papatasi* suitable areas. As such, implementing targeted control and surveillance measures in areas identified as having a very high CL vector suitability (<0.9 suitability) would include a substantial proportion (72% for GHS and 89% for GPW) of the population at risk while encompassing only 3.42% of the country's total geographic area. Therefore, high impact public health interventions could be achieved within a reduced spatial target, thus maximizing the efficient use of resources.

Important drivers of emergence, re-emergence and/or expansion of vector-borne diseases include global travel, large scale human migrations and climate change. The former two are relevant in the context of epidemic/pandemic opportunities and instability-related migrations, while the latter may facilitate the extension of the spatio-temporal availability of favourable conditions for vector survival and reproduction. Jordan's geographical location has resulted in it being a main destination for refugees fleeing violence in Syria. While endemic in both Jordan and Syria, CL has been reported considerably more frequently in Syria with a significant surge since the onset of political terror and the civil war [3,27,29,85], and the risk of a CL outbreak

| Month     | Mean change for equal weights | Mean change for linear membership function | P-value |
|-----------|-------------------------------|--|---------|
| April     | 0 ± 0.11                      | $0.07 \pm 0.12$                            | < 0.01  |
| May       | $0.05 \pm 0.06$               | $0.06 \pm 0.1$                             | > 0.05  |
| June      | $0.07 \pm 0.14$               | 0 ± 0.12                                   | < 0.01  |
| July      | $0.02 \pm 0.09$               | $0.12 \pm 0.15$                            | < 0.01  |
| August    | $0.05 \pm 0.17$               | $0.12 \pm 0.2$                             | < 0.01  |
| September | 0.18 ± 0.11                   | $0.18 \pm 0.18$                            | > 0.05  |
| Overall   | 0.06 ± 0.13                   | 0.09 ± 0.16                                | < 0.01  |

Table 9. Sensitivity analysis results. Mean change in values (n = 96) between original multicriteria decision analysis outputs compared to new maps with either equal weights for all predictor variables, or assuming linear membership functions for all predictor variables.

https://doi.org/10.1371/journal.pntd.0008852.t009



Fig 7. Population at risk of *Phlebotomus papatasi* exposure between April and September using different suitability cut-off points from moderate (0.5) to very high (0.9) suitability calculated using (top) the Global Population of the World (GPW) grid and (bottom) the European Commission's Global Human Settlement (GHS) population grid.

in Jordan may increase if potential CL-carriers are living in CL vector-suitable areas along with a large, possibly immuno-naïve, population.

The use of *P. papatasi* risk maps was a useful proxy for exploring risk of CL instead of directly using CL epidemiological data for several reasons. Firstly, CL reporting is carried out at governorate level so using epidemiological data would result in risk maps at a very coarse spatial resolution. Secondly, the possibly severe under-reporting of CL [26], which may be due to treatment-seeking socio-economical barriers or behaviours, might introduce bias to any map based on epidemiological data. Jordan's MoH provides free healthcare to the Jordanian population and registered Syrian refugees. However, for those displaced Syrians who remain

| Table 10.  | Population at risk res  | ults. Quantification of | of (i) areas suitable for | r Phlebotomus papatasi     | occurrence in relation t  | o the whole country; (ii  | ) the population at risk |
|------------|-------------------------|-------------------------|---------------------------|----------------------------|---------------------------|---------------------------|--------------------------|
| of cutanec | ous leishmaniasis using | different population g  | grids; and (iii) the pop  | pulation at risk in relati | on to the total populatio | n, using different suitab | ility cut-off values.    |

| Suitability cut-<br>off | % of Jordan within suitability range | PAR (GHS) | % of total Jordan population at risk<br>(GHS) | PAR (GPW) | % of total Jordan population at risk<br>(GPW) |
|-------------------------|--------------------------------------|-----------|---|-----------|---|
| 0.5                     | 57.44                                | 610,100   | 13.26   | 7,535,141 | 99.98   |
| 0.6                     | 46.77                                | 598,801   | 13.01   | 7,506,769 | 99.60   |
| 0.7                     | 34.92                                | 550,408   | 11.96   | 7,422,532 | 98.49   |
| 0.8                     | 10.35                                | 480,652   | 10.45   | 7,217,799 | 95.77   |
| 0.9                     | 3.42                                 | 443,269   | 9.63  | 6,749,171 | 89.55   |

https://doi.org/10.1371/journal.pntd.0008852.t010

unregistered or do not possess any official documentation, access to adequate healthcare is challenging and dependent on a network of non-governmental organisations. Moreover, increased pressure on resources leading to shortage of medical staff and supplies has resulted in tensions between the Jordanian and Syrian populations, some of whom have resorted to seeking medical care in private clinics [86]. Because of inadequate and challenging access to healthcare, people might seek treatment only for severe or facial lesions as CL is generally known to be self-curing [32,87,88]. Therefore, the migration of Syrian displaced populations, including possible CL carriers, may drastically affect the national epidemiology but might not be captured as a consequence of the ongoing challenges to adequate healthcare access, underreporting and passive surveillance [26,29,30,85,86,89]. The model outputs presented in this study could therefore, be used to highlight priority target areas to minimize CL burden. However, factors beyond the geographical distribution and determinants of disease such as financial cost, perceived risk, and existing policies and priorities in allocation of resources [35], need to be considered in public health decision-making.

#### Risk maps and MCDA for surveillance and control of vector-borne diseases

Spatial risk models for CL, as with other vector borne diseases, are valuable in characterizing the entomological risk and thus, exposure to a disease vector. However, they are only one tool within a larger and more complex decision-making framework for the prevention and control of vector-borne diseases. Despite their limitations they are inexpensive, and in the absence of detailed field-based surveillance information, MCDA provides preliminary estimates of risk that, although imperfect, can be used for the initial stages of planning a control program or setting priorities [35]. In addition, this work can be easily updated with new information, which is an important advantage given the current changing global climate and population dynamics influenced by growth and migration.

Spatial MCDA has been used to set the level of risk for Rift Valley Fever (RVF) by incorporating environmental drivers, expert opinions and uncertainties to create continent-wide estimates of RVF activity [49]. Spatial MCDA has also been used to model habitat suitability for highly pathogenic avian influenza in south-east Asia [55] and low pathogenic avian influenza in Argentina [90]. Such work can be taken a step further by comparing two risk models, which was carried out by Sarkar *et al.* [50] to understand Chagas disease risk in Texas [51] where one model addressed vector distribution while a second model addressed parasite presence, allowing a more robust understanding of overall risk. MCDA could be applied to evaluate a set of vector control strategies to identify the best alternative among all possible options (e.g. pesticide spraying, promotion of protective behaviours such as using repellent or bed nets, or larvicide programmes). Rakotomanana *et al.* [50] used spatial MCDA to assess malaria risk in the highlands of Madagascar to identify the most effective use of resources for vector control. Decision-making criteria included elevation, population density, time since last indoor spraying, distance from rice fields and district level surface area of rice fields. The results from the spatial MCDA were then used to define priority zones to carry out indoor spraying [50].

#### MCDA advantages and limitations

This study exemplifies how MCDA can be a powerful tool that can be applied to address complex decisions in data-scarce settings. MCDA is advantageous in that it can incorporate different and even conflicting views of several stakeholders and so criteria are chosen based on these values along with any potential constraints (e.g. reduction of disease-related morbidity/mortality, environmental impact, costs, duration of effectivity). This approach allows the integration of criteria that traditionally would not have been able to be considered simultaneously, and to rank alternative control strategies based on specified evaluation criteria.

However, there are limitations that need to be considered when interpreting these results. The most important consideration is that the quality of the model outputs will depend on the quality of the input data [33] and that the model might be subject to a degree of uncertainty that can be input-related linked to the source predictor data layers, and/or process-related resulting from data transformations and spatial interpolation algorithms [75]. The climatic data included in the model showed relatively good agreement with Jordan weather station data except for wind speed and relative humidity. This could have been due to the use of different measuring methodologies to capture wind speed as measurements depend greatly on the height at which they are measured, roughness of the ground and surrounding structures like buildings or trees [91]. Wind speed was a new variable only recently included within World-Clim V.2 (2017), so it might require time to optimize outputs and adjust errors. CliMond produced relative humidity maps at an arc 10' spatial resolution, which might be too coarse to agree with weather station data. Population grids are highly variable and particularly inaccurate for low and middle income countries [75]. This is due to limited up-to-date demographic data as population censuses are usually carried out every ten years [75]. The marked difference between GHS and GPW population grids used in this study may be due to the use of different methodologies. An important difference includes the use of United Nations (UN)-adjusted population data in GPW, whereby the use of national statistic census data are adjusted to match official UN population estimates [92].

The second important consideration to keep in mind when interpreting MCDA outputs is the subjectivity involved in the methodology especially in (i) the selection of predictor variables; (ii) the definition of the relationship between predictor variable and outcome suitability using fuzzy membership functions; and finally (iii) the allocation of weights. An individual's perspective on these subjects is expected to be influenced by personal experience and expertise although there are ways to address and ameliorate the possible bias and uncertainty that an individual may introduce. One option involves group discussions with experts and stakeholders to reach a group consensus. An alternative option involves the use of the scientific literature and allocating weights based on the number of times a variable has been reported along with its statistical significance [49,93] This study adopted a mixture between these two methods. Expert elicitation and group consensus was not possible due to the low CR of each of the returned questionnaires. The quantitative inclusion of some variables such as vegetation (NDVI), P. obesus distribution, wind speed or human settlement, are novel in P. papatasi ecology research which was reflected in the experts' decreased confidence scores for these variables. For this reason the "hit-based" method applied by Stevens et al., 2013 [93] or Clements et al., 2006 [49] would not have been fully suitable either as important variables that influence *P. papatasi* occurrence are under-represented in the published literature. The restricted use of these variables in previous studies could be due to their limited availability or technical requirements associated with their use: WorldClim has only just recently included wind speed within its data repository; NDVI and human settlement maps require analysis of remote sensing satellite data; and a proxy was required and produced for *P. obesus* distribution. This highlights how data availability and resolution are important limitations for MCDA outputs. In addition, the range of predictors included in the model (e.g. vegetation (NDVI), P. obesus distribution, wind speed) means that to be effective, any expertise required would be from very diverse fields, therefore several panels of experts would be needed including botanists, zoologists, and entomologists, and therefore it was decided that these factors justified using a literature review across fields, and judgement by the first author.

Despite concerns associated with potential error due to the use of subjective risk factor weighting, this study showed that altering weights had a significantly lower effect on the final suitability estimate compared to altering the membership functions. These results are in line with findings by Stevens *et al.* (2013) [93] and Clements *et al.* (2006) [49], therefore emphasizing the importance of carefully considering and understanding the implications of the use of different membership functions.

A wide range of metrics are available to compare the veracity of model predictions with observations, whether based on a totally independent test data set, or on resampled observations within the training set, as in the case of cross-validation or bootstrapping. However, by their very nature these validation techniques assume the availability of both disease or vector presence and absence data; an inherent limitation in knowledge-driven modelling. However, model validation is inextricably linked with model uncertainty which can be either epistemic (e.g. measurement error, natural variability, model uncertainty) or stochastic (i.e. uncertainty due to inherent variability in the underlying biological processes [94,95]. Although uncertainty resulting from data errors can be substantial, of equal importance is the uncertainty resulting from missing predictors due to incomplete knowledge of the organism being modelled or because the spatial data for the predictor are unavailable [96]. This study has reduced this form of uncertainty and set itself apart from previous studies investigating CL vectors by including a set of carefully selected variables using evidence-based biological rationale rather than data availability. Other predictive studies on CL vector distribution have mostly included readily available variables such as the WorldClim bioclimatic variables [41-43,45,46], which are easily accessible and provide a range of resolutions; the highest being 1 km<sup>2</sup>. Despite the quality of WorldClim data, of the available 19 bioclimatic variables, 11 are temperature-related and the remaining 8 are precipitation-related. Therefore, the diversity of environmental variables included in previous predictive models is limited.

Estimates of uncertainty are generally only generated or considered at the end of the modelling process despite that, whenever possible, the concept should be incorporated at all stages [95]. In this respect MCDA may be considered to have an advantage over other modelling methods as the use of fuzzy logic and fuzzy membership functions inherently incorporates uncertainty about the relationships between predictors and outcome within the modelling process.

# Conclusions

Habitat suitability for the main CL vector *P. papatasi* in Jordan is heterogeneous over space and time. Regions in the north-west are most suitable for *P. papatasi* occurrence and the suitability range increases with the progression of the warmer summer months. Suitable areas for CL vector occurrence coincide with highly populated parts of the country. Geographically targeting mitigation strategies can benefit a large proportion of the population at risk of CL, and thus promote resource-efficient interventions. This is, however, one component within the complex requirements for the implementation of a comprehensive national strategy for CL prevention and control in Jordan. In addition to the epidemiological elements, other socioeconomic and logistical constraints will determine the success of a national vector control program.

# Acknowledgments

The authors would firstly, like to thank to Dr Stephan Kern from the Integrated Climate Data Center (University of Hamburg), Dr Niels Batjes from the International Soil Reference and

Information Centre, and Dr Raphaelle Metras from the London School of Hygiene and Tropical Medicine for their advice and data contributions.

# **Author Contributions**

Conceptualization: Emi A. Takahashi, Rami Mukbel, Javier Guitian, Kim B. Stevens.

Data curation: Emi A. Takahashi.

Formal analysis: Emi A. Takahashi, Kim B. Stevens.

Investigation: Emi A. Takahashi.

Methodology: Emi A. Takahashi, Kim B. Stevens.

Project administration: Javier Guitian.

Resources: Emi A. Takahashi, Lina Masoud.

Supervision: Kim B. Stevens.

Visualization: Emi A. Takahashi.

Writing - original draft: Emi A. Takahashi.

Writing – review & editing: Emi A. Takahashi, Lina Masoud, Rami Mukbel, Javier Guitian, Kim B. Stevens.

#### References

- 1. WHO. Neglected tropical diseases [16 June 2017]. Available from: http://www.who.int/neglected\_diseases/diseases/en/.
- Alvar J, Vélez ID, Bern C, Herrero M, Desjeux P, Cano J, et al. Leishmaniasis worldwide and global estimates of its incidence. PloS one. 2012; 7(5):e35671. https://doi.org/10.1371/journal.pone.0035671 PMID: 22693548
- Salam N, Al-Shaqha WM, Azzi A. Leishmaniasis in the middle East: incidence and epidemiology. PLoS Negl Trop Dis. 2014; 8(10):e3208. https://doi.org/10.1371/journal.pntd.0003208 PMID: 25275483; PubMed Central PMCID: PMC4183486.
- Murray HW, Berman JD, Davies CR, Saravia NG. Advances in leishmaniasis. Lancet. 2005; 366 (9496):1561–77. https://doi.org/10.1016/S0140-6736(05)67629-5 PMID: 16257344.
- Akhoundi M, Kuhls K, Cannet A, Votypka J, Marty P, Delaunay P, et al. A Historical Overview of the Classification, Evolution, and Dispersion of Leishmania Parasites and Sandflies. PLoS Negl Trop Dis. 2016; 10(3):e0004349. Epub 2016/03/05. https://doi.org/10.1371/journal.pntd.0004349 PMID: 26937644; PubMed Central PMCID: PMC4777430.
- Gramiccia M, Gradoni L. The current status of zoonotic leishmaniases and approaches to disease control. Int J Parasitol. 2005; 35(11–12):1169–80. https://doi.org/10.1016/j.ijpara.2005.07.001 PMID: 16162348.
- Reithinger R, Dujardin JC, Louzir H, Pirmez C, Alexander B, Brooker S. Cutaneous leishmaniasis. The Lancet Infectious diseases. 2007; 7(9):581–96. Epub 2007/08/24. https://doi.org/10.1016/S1473-3099(07)70209-8 PMID: 17714672.
- Saliba EK, Disi AM, Ayed RE, Saleh N, al-Younes H, Oumeish O, et al. Rodents as reservoir hosts of cutaneous leishmaniasis in Jordan. Annals of tropical medicine and parasitology. 1994; 88(6):617–22. Epub 1994/12/01. https://doi.org/10.1080/00034983.1994.11812912 PMID: 7893175.
- Schlein Y, Warburg A, Schnur LF, Le Blancq SM, Gunders AE. Leishmaniasis in Israel: reservoir hosts, sandfly vectors and leishmanial strains in the Negev, Central Arava and along the Dead Sea. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1984; 78(4):480–4. Epub 1984/ 01/01. https://doi.org/10.1016/0035-9203(84)90067-1 PMID: 6385358.
- Saliba EK, Higashi GI, Yates JA, Oumeish OY. Cutaneous leishmaniasis in Jordan: biochemical identification of human and Psammomys obesus isolates as Leishmania major. Annals of tropical medicine and parasitology. 1988; 82(1):21–5. Epub 1988/02/01. https://doi.org/10.1080/00034983.1988. 11812204 PMID: 3041929.

- Fichet-Calvet E, Jomaa I, Ben Ismail R, Ashford RW. Leishmania major infection in the fat sand rat Psammomys obesus in Tunisia: interaction of host and parasite populations. Annals of tropical medicine and parasitology. 2003; 97(6):593–603. Epub 2003/09/27. https://doi.org/10.1179/ 000349803225001517 PMID: 14511558.
- 12. Annajar BB. Epidemiology of cutaneous leishmaniasis in Lybia: Keele University; 1999.
- Samy AM, Campbell LP, Peterson AT. Leishmaniasis transmission: distribution and coarse-resolution ecology of two vectors and two parasites in Egypt. Revista da Sociedade Brasileira de Medicina Tropical. 2014; 47(1):57–62. Epub 2014/03/08. https://doi.org/10.1590/0037-8682-0189-2013 PMID: 24603738.
- Wasserberg G, Yarom I, Warburg A. Seasonal abundance patterns of the sandfly Phlebotomus papatasi in climatically distinct foci of cutaneous leishmaniasis in Israeli deserts. Medical and veterinary entomology. 2003; 17(4):452–6. Epub 2003/12/04. https://doi.org/10.1111/j.1365-2915.2003.00461.x PMID: 14651661.
- Janini R, Saliba E, Khoury S, Oumeish O, Adwan S, Kamhawi S. Incrimination of Phlebotomus papatasi as vector of Leishmania major in the southern Jordan Valley. Medical and veterinary entomology. 1995; 9(4):420–2. Epub 1995/10/01. https://doi.org/10.1111/j.1365-2915.1995.tb00016.x PMID: 8541595.
- Saliba E, Saleh N, Bisharat Z, Oumeish O, Khoury S, Gramiccia M, et al. Cutaneous leishmaniasis due to Leishmania tropica in Jordan. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1993; 87(6):633. Epub 1993/11/01. https://doi.org/10.1016/0035-9203(93)90267-t PMID: 8296360.
- Saliba EK, Saleh N, Oumeish OY, Khoury S, Bisharat Z, al-Ouran R. The endemicity of Leishmania tropica (zymodeme MON-137) in the Eira-Yarqa area of Salt District, Jordan. Annals of tropical medicine and parasitology. 1997; 91(5):453–9. Epub 1997/07/01. PMID: 9329981.
- Kamhawi S, Abdel-Hafez SK, Arbagi A. A new focus of cutaneous leishmaniasis caused by Leishmania tropica in northern Jordan. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1995; 89(3):255–7. Epub 1995/05/01. https://doi.org/10.1016/0035-9203(95)90526-x PMID: 7660425.
- Mosleh IM, Geith E, Schonian G, Kanani KA. Two recent but temporally distinct outbreaks of cutaneous leishmaniasis among foreign workers in the Dead-Sea area of Jordan. Annals of tropical medicine and parasitology. 2009; 103(5):393–400. Epub 2009/07/09. <u>https://doi.org/10.1179/ 136485909X451735 PMID: 19583910.</u>
- Jumaian N, Kamhawi SA, Halalsheh M, Abdel-Hafez SK. Short report: outbreak of cutaneous leishmaniasis in a nonimmune population of soldiers in Wadi Araba, Jordan. Am J Trop Med Hyg. 1998; 58 (2):160–2. https://doi.org/10.4269/ajtmh.1998.58.160 PMID: 9502598.
- Arbaji AK, Gradoni L, Gramiccia M. Leishmanin skin test survey in a focus of high endemicity of Leishmania major in Jordan. Acta tropica. 1993; 54(1):77–9. Epub 1993/06/01. <u>https://doi.org/10.1016/0001-706x(93)90070-r PMID: 8103629</u>.
- 22. Mosleh IM, Shonian G, Geith E, Al-Jawabreh A, Natsheh L. The Jordanian Mid Jordan Valley is a classic focus of Leishmania major as revealed by RFLP of 56 isolates and 173 ITS-1-PCR-positive clinical samples. Experimental parasitology. 2015; 148:81–5. Epub 2014/12/03. <u>https://doi.org/10.1016/j.exppara.2014.11.006 PMID: 25450773</u>.
- Nimri L, Soubani R, Gramiccia M. Leishmania species and zymodemes isolated from endemic areas of cutaneous leishmaniasis in Jordan. Kinetoplastid biology and disease. 2002; 1(1):7. Epub 2002/12/ 11. https://doi.org/10.1186/1475-9292-1-7 PMID: 12473179; PubMed Central PMCID: PMC149425.
- Janini R, Saliba E, Kamhawi S. Species composition of sand flies and population dynamics of Phlebotomus papatasi (Diptera: Psychodidae) in the southern Jordan Valley, an endemic focus of cutaneous leishmaniasis. Journal of medical entomology. 1995; 32(6):822–6. Epub 1995/11/01. <u>https://doi.org/</u> 10.1093/jmedent/32.6.822 PMID: 8551505.
- 25. Kamhawi S, Arbagi A, Adwan S, Rida M. Environmental manipulation in the control of a zoonotic cutaneous leishmaniasis focus. Arch Inst Pasteur Tunis. 1993; 70(3–4):383–90. PMID: 7802493.
- Mosleh IM, Geith E, Natsheh L, Abdul-Dayem M, Abotteen N. Cutaneous leishmaniasis in the Jordanian side of the Jordan Valley: severe under-reporting and consequences on public health management. Trop Med Int Health. 2008; 13(6):855–60. https://doi.org/10.1111/j.1365-3156.2008.02063.x PMID: 18363585.
- 27. Berry I, Berrang-Ford L. Leishmaniasis, conflict, and political terror: a spatio-temporal analysis. Social Science & Medicine. 2016; 167:140–9.
- 28. UNHCR. Syria Regional Refugee Response—Regional Overview 2017 [cited 2016 03/02/2017]. Available from: http://data.unhcr.org/syrianrefugees/regional.php.

- Saroufim M, Charafeddine K, Issa G, Khalifeh H, Habib RH, Berry A, et al. Ongoing epidemic of cutaneous leishmaniasis among Syrian refugees, Lebanon. Emerg Infect Dis. 2014; 20(10):1712–5. https://doi.org/10.3201/eid2010.140288 PMID: 25279543; PubMed Central PMCID: PMC4193275.
- Koltas IS, Eroglu F, Alabaz D, Uzun S. The emergence of Leishmania major and Leishmania donovani in southern Turkey. Trans R Soc Trop Med Hyg. 2014; 108(3):154–8. <u>https://doi.org/10.1093/trstmh/</u> trt119 PMID: 24449479.
- Rowland M, Munir A, Durrani N, Noyes H, Reyburn H. An outbreak of cutaneous leishmaniasis in an Afghan refugee settlement in north-west Pakistan. Trans R Soc Trop Med Hyg. 1999; 93(2):133–6. https://doi.org/10.1016/s0035-9203(99)90285-7 PMID: 10450434.
- WHO. Framework for action on cutaneous leishmaniasis in the Eastern Mediterranean Region 2014– 2018. World Health Organization. Regional Office for the Eastern Mediterranean, 2014.
- Eisen L, Eisen RJ. Using geographic information systems and decision support systems for the prediction, prevention, and control of vector-borne diseases. Annual review of entomology. 2011; 56:41–61. https://doi.org/10.1146/annurev-ento-120709-144847 PMID: 20868280
- **34.** Stevens KB, Pfeiffer DU. Spatial modelling of disease using data-and knowledge-driven approaches. Spatial and spatio-temporal epidemiology. 2011; 2(3):125–33. https://doi.org/10.1016/j.sste.2011.07. 007 PMID: 22748172
- **35.** Hongoh V, Hoen AG, Aenishaenslin C, Waaub J-P, Bélanger D, Michel P. Spatially explicit multi-criteria decision analysis for managing vector-borne diseases. Int J Health Geog. 2011; 10(1):70. <u>https://</u> doi.org/10.1186/1476-072X-10-70 PMID: 22206355
- Baron SD, Morillas-Marquez F, Morales-Yuste M, Diaz-Saez V, Gallego M, Molina R, et al. Predicting the risk of an endemic focus of Leishmania tropica becoming established in South-Western Europe through the presence of its main vector, Phlebotomus sergenti Parrot, 1917. Parasitology. 2013; 140 (11):1413–21. Epub 2013/08/24. https://doi.org/10.1017/S0031182013000942 PMID: 23965821.
- Galvez R, Descalzo MA, Guerrero I, Miro G, Molina R. Mapping the current distribution and predicted spread of the leishmaniosis sand fly vector in the madrid region (Spain) based on environmental variables and expected climate change. Vector Borne Zoonotic Dis. 2011; 11(7):799–806. Epub 2011/03/ 23. https://doi.org/10.1089/vbz.2010.0109 PMID: 21417927.
- Cross ER, Newcomb WW, Tucker CJ. Use of weather data and remote sensing to predict the geographic and seasonal distribution of Phlebotomus papatasi in southwest Asia. Am J Trop Med Hyg. 1996; 54(5):530–6. Epub 1996/05/01. https://doi.org/10.4269/ajtmh.1996.54.530 PMID: 8644911.
- Kassem HA, Siri J, Kamal HA, Wilson ML. Environmental factors underlying spatial patterns of sand flies (Diptera: Psychodidae) associated with leishmaniasis in southern Sinai, Egypt. Acta tropica. 2012; 123(1):8–15. Epub 2012/03/14. <u>https://doi.org/10.1016/j.actatropica.2012.02.067</u> PMID: 22410540.
- Kahime K, Boussaa S, El Mzabi A, Boumezzough A. Spatial relations among environmental factors and phlebotomine sand fly populations (Diptera: Psychodidae) in central and southern Morocco. J Vector Ecol. 2015; 40(2):342–54. Epub 2015/11/28. https://doi.org/10.1111/jvec.12173 PMID: 26611970.
- Colacicco-Mayhugh MG, Masuoka PM, Grieco JP. Ecological niche model of Phlebotomus alexandri and P. papatasi (Diptera: Psychodidae) in the Middle East. International journal of health geographics. 2010; 9:2. Epub 2010/01/22. https://doi.org/10.1186/1476-072X-9-2 PMID: 20089198; PubMed Central PMCID: PMC2823717.
- Ferro C, Lopez M, Fuya P, Lugo L, Cordovez JM, Gonzalez C. Spatial Distribution of Sand Fly Vectors and Eco-Epidemiology of Cutaneous Leishmaniasis Transmission in Colombia. PLoS One. 2015; 10 (10):e0139391. Epub 2015/10/03. https://doi.org/10.1371/journal.pone.0139391 PMID: 26431546; PubMed Central PMCID: PMC4592259.
- Samy AM, Annajar BB, Dokhan MR, Boussaa S, Peterson AT. Coarse-resolution Ecology of Etiological Agent, Vector, and Reservoirs of Zoonotic Cutaneous Leishmaniasis in Libya. PLoS Negl Trop Dis. 2016; 10(2):e0004381. https://doi.org/10.1371/journal.pntd.0004381 PMID: 26863317; PubMed Central PMCID: PMC4749236.
- 44. Hanafi-Bojd AA, Yaghoobi-Ershadi MR, Haghdoost AA, Akhavan AA, Rassi Y, Karimi A, et al. Modeling the Distribution of Cutaneous Leishmaniasis Vectors (Psychodidae: Phlebotominae) in Iran: A Potential Transmission in Disease Prone Areas. Journal of medical entomology. 2015; 52(4):557–65. Epub 2015/09/04. https://doi.org/10.1093/jme/tjv058 PMID: 26335462.
- 45. Abedi-Astaneh F, Akhavan AA, Shirzadi MR, Rassi Y, Yaghoobi-Ershadi MR, Hanafi-Bojd AA, et al. Species diversity of sand flies and ecological niche model of Phlebotomus papatasi in central Iran. Acta Trop. 2015; 149:246–53. https://doi.org/10.1016/j.actatropica.2015.05.030 PMID: 26071647.

- Abdel-Dayem MS, Annajar BB, Hanafi HA, Obenauer PJ. The potential distribution of Phlebotomus papatasi (Diptera: Psychodidae) in Libya based on ecological niche model. J Med Entomol. 2012; 49 (3):739–45. Epub 2012/06/12. https://doi.org/10.1603/me11225 PMID: 22679884.
- 47. Saaty TL. The Analytic Hierarchy Process. New York, NY: McGraw-Hill; 1980.
- 48. Malczewski J. GIS and Multicriteria Decision Analysis. New York: John Wiley & Sons, Inc; 1999.
- **49.** Clements AC, Pfeiffer DU, Martin V. Application of knowledge-driven spatial modelling approaches and uncertainty management to a study of Rift Valley fever in Africa. International Journal of Health Geographics. 2006; 5(1):57. https://doi.org/10.1186/1476-072X-5-57 PMID: 17156467
- 50. Rakotomanana F, Randremanana RV, Rabarijaona LP, Duchemin JB, Ratovonjato J, Ariey F, et al. Determining areas that require indoor insecticide spraying using Multi Criteria Evaluation, a decisionsupport tool for malaria vector control programmes in the Central Highlands of Madagascar. Int J Health Geog. 2007; 6(1):2.
- Sarkar S, Strutz SE, Frank DM, Rivaldi CL, Sissel B, Sánchez–Cordero V. Chagas disease risk in Texas. PLoS Neglected Tropical Diseases. 2010; 4(10):e836. https://doi.org/10.1371/journal.pntd. 0000836 PMID: 20957148
- Rousseau R, McGrath G, McMahon BJ, Vanwambeke SO. Multi-criteria Decision Analysis to Model Ixodes ricinus Habitat Suitability. EcoHealth. 2017:1–12. <u>https://doi.org/10.1007/s10393-017-1220-6</u> PMID: 28194584
- Alimi TO, Fuller DO, Herrera SV, Arevalo-Herrera M, Quinones ML, Stoler JB, et al. A multi-criteria decision analysis approach to assessing malaria risk in northern South America. BMC Public Health. 2016; 16:221. <u>https://doi.org/10.1186/s12889-016-2902-7</u> PMID: <u>26940004</u>; PubMed Central PMCID: PMC4778356.
- Lane RP, Abdel-Hafez S, Kamhawi S. The distribution of phlebotomine sandflies in the principal ecological zones of Jordan. Medical and veterinary entomology. 1988; 2:237–46. <u>https://doi.org/10.1111/j. 1365-2915.1988.tb00190.x PMID: 2980180</u>
- 55. Stevens K, Gilbert M, Pfeiffer D. Modeling habitat suitability for occurrence of highly pathogenic avian influenza virus H5N1 in domestic poultry in Asia: a spatial multicriteria decision analysis approach. Spat Spatiotemporal Epidemiol. 2013; 4:1–14. https://doi.org/10.1016/j.sste.2012.11.002 PMID: 23481249
- Pfeiffer DU, Robinson TP, Stevenson M, Stevens KB, Rogers DJ, Clements ACA. Spatial risk assessment and management of disease. Spatial Analysis in Epidemiology. Oxford: Oxford University Press; 2008.
- Mollalo A, Khodabandehloo E. Zoonotic cutaneous leishmaniasis in northeastern Iran: a GIS-based spatio-temporal multi-criteria decision-making approach. Epidemiology & Infection. 2016; 144 (10):2217–29. https://doi.org/10.1017/S0950268816000224 PMID: 26931076
- 58. The World Bank 2016 [17/07/2017]. Available from: http://data.worldbank.org/.
- 59. Al-Eisawi D. Conservation of natural ecosystems in Jordan. Pak J Bot. 2012; 44:95-9.
- 60. Zadeh LA. Fuzzy sets. Information and control. 1965; 8(3):338-53.
- Saaty TL. A scaling method for priorities in hierarchical structures. Journal of mathematical psychology. 1977; 15(3):234–81.
- 62. R. Hijmans ea. GADM database of Global Administrative Areas. 2.8 ed2015.
- 63. ESRI. ArcGIS 10.4.1. 10. Redlands, CA: Environmental Systems Research Institute2016.
- 64. Fick SE, Hijmans RJ. WorldClim 2: new 1-km spatial resolution climate surfaces for global land areas. International Journal of Climatology. 2017.
- Kriticos DJ, Webber BL, Leriche A, Ota N, Macadam I, Bathols J, et al. CliMond: global high-resolution historical and future scenario climate surfaces for bioclimatic modelling. Methods in Ecology and Evolution. 2012; 3(1):53–64.
- 66. RStudioTeam. Studio: Integrated Development for R. RStudio, Inc., Boston, MA 2016.
- FICHET-CALVET E, JOMAA I ZAAFOURIJ B, ASHFORD R, BEN-ISMAIL R, DELATTRE P The spatio-temporal distribution of a rodent reservoir host of cutaneous leishmaniasis. J Appl Ecol. 2000; 37:603–15.
- Tchabovsky A, Krasnov B. Spatial distribution of Psammomys obesus (Rodentia, Gerbillinae) in relation to vegetation in the Negev desert of Israel. Mammalia. 2002; 66(3):361–8.
- 69. Kuznetsova TA, Kam M, Khokhlova IS, Kostina NV, Dobrovolskaya TG, Umarov MM, et al. Desert gerbils affect bacterial composition of soil. Microbial ecology. 2013; 66(4):940–9. Epub 2013/07/17. https://doi.org/10.1007/s00248-013-0263-7 PMID: 23857378.



- 70. El-Bana MI. Effects of the abandonment of the burrowing mounds of fat sand rat (Psammomys obesus cretzschamar 1828) on vegetation and soil surface attributes along the coastal dunes of North Sinai, Egypt. Journal of arid environments. 2009; 73(9):821–7.
- El-Ghani MA, Soliman A, El-Fattah RA. Spatial distribution and soil characteristics of the vegetation associated with common succulent plants in Egypt. Turkish Journal of Botany. 2014; 38(3):550–65.
- 72. Batjes NH RZaA-AR. Soil data derived from SOTER for studies of carbon stocks and change in Jordan GEFSOC Project: ISRIC—World Soil Information, Wageningen, 2003.
- 73. Malczewski J. On the use of weighted linear combination method in GIS: common and best practice approaches. Trans GIS. 2000; 4:5–20.
- 74. TerrSet. Geospatial Monitoring and Modeling Software. Clark Labs, Clark University 950 Main St., Worcester MA 01610 USA2015.
- 75. Tatem AJ, Campiz N, Gething PW, Snow RW, Linard C. The effects of spatial population dataset choice on estimates of population at risk of disease. Population Health Metrics. 2011; 9(1):4. <u>https:// doi.org/10.1186/1478-7954-9-4 PMID: 21299885</u>
- 76. Boussaa S, Guernaoui S, Pesson B, Boumezzough A. Seasonal fluctuations of phlebotomine sand fly populations (Diptera: Psychodidae) in the urban area of Marrakech, Morocco. Acta tropica. 2005; 95 (2):86–91. <u>https://doi.org/10.1016/j.actatropica.2005.05.002</u> PMID: <u>15985259</u>
- EI-Shazly MM, Soliman MM, Zayed A. Seasonal abundance, number of annual generations, and effect of an entomopathogenic fungus on Phlebotomus papatasi (Diptera: Psychodidae). Environmental entomology. 2012; 41(1):11–9. https://doi.org/10.1603/EN11109 PMID: 22525055
- 78. Emami MM, Yazdi M. Entomological survey of phlebotomine sand flies (Diptera: Psychodidae) in a focus of visceral leishmaniasis in central Iran. Journal of vector borne diseases. 2008; 45(1):38. PMID: 18399315
- Orshan L, Szekely D, Khalfa Z, Bitton S. Distribution and seasonality of Phlebotomus sand flies in cutaneous leishmaniasis foci, Judean Desert, Israel. J Med Entomol. 2010; 47(3):319–28. Epub 2010/ 05/26. https://doi.org/10.1603/me09096 PMID: 20496578.
- Muller GC, Revay EE, Schlein Y. Relative attraction of the sand fly Phlebotomus papatasi to local flowering plants in the Dead Sea region. J Vector Ecol. 2011; 36 Suppl 1:S187–94. Epub 2011/03/05. https://doi.org/10.1111/j.1948-7134.2011.00130.x PMID: 21366774.
- Sawalha SS, Shtayeh MS, Khanfar HM, Warburg A, Abdeen ZA. Phlebotomine sand flies (Diptera: Psychodidae) of the Palestinian West Bank: potential vectors of leishmaniasis. J Med Entomol. 2003; 40(3):321–8. Epub 2003/08/29. https://doi.org/10.1603/0022-2585-40.3.321 PMID: 12943111.
- 82. Theodor O. On the relation of Phlebotomus papatasii to the temperature and humidity of the environment. Bulletin of Entomological Research. 1936; 27(04):653–71.
- Kasap OE, Alten B. Laboratory estimation of degree-day developmental requirements of Phlebotomus papatasi (Diptera: Psychodidae). J Vector Ecol. 2005; 30(2):328–33. PMID: 16599172.
- Anis E, Leventhal A, Elkana Y, Wilamowski A, Pener H. Cutaneous leishmaniasis in Israel in the era of changing environment. Public health reviews. 2001; 29(1):37–47. Epub 2002/01/10. PMID: <u>11780715</u>.
- Fawcett JM, Hay RJ. Cutaneous leishmaniasis and human conflict. Acta Derm Venereol. 2015; 95 (1):3–4. Epub 2014/11/05. https://doi.org/10.2340/00015555-1994 PMID: 25363408.
- 86. REACH. Access to health care and tensions in Jordanian communities hosting Syrian refugees. 2014.
- Ameen M. Cutaneous leishmaniasis: advances in disease pathogenesis, diagnostics and therapeutics. Clinical and experimental dermatology. 2010; 35(7):699–705. https://doi.org/10.1111/j.1365-2230.2010.03851.x PMID: 20831602
- Karimkhani C, Wanga V, Coffeng LE, Naghavi P, Dellavalle RP, Naghavi M. Global burden of cutaneous leishmaniasis: a cross-sectional analysis from the Global Burden of Disease Study 2013. The Lancet Infectious Diseases. 2016; 16(5):584–91. <u>https://doi.org/10.1016/S1473-3099(16)00003-7</u> PMID: 26879176
- UNHCR. Syria Regional Refugee Response 2017 [26/07/2017]. Available from: <u>http://www.data.unhcr.org/syrianrefugees/country.php?id=107</u>.
- 90. La Sala LF, Burgos JM, Blanco DE, Stevens KB, Fernández AR, Capobianco G, et al. Spatial modelling for low pathogenicity avian influenza virus at the interface of wild birds and backyard poultry. Transboundary and Emerging Diseases. 2019; 66(4):1493–505. https://doi.org/10.1111/tbed.13136 PMID: 30698918
- 91. MetOffice. How we measure wind 2015. Available from: http://www.metoffice.gov.uk/guide/weather/ observations-guide/how-we-measure-wind.
- Deichmann U, Balk D, Yetman G. Transforming population data for interdisciplinary usages: from census to grid. Book Transforming population data for interdisciplinary usages: from census to grid. 2001.

- 93. Stevens KB, Gilbert M, Pfeiffer DU. Modeling habitat suitability for occurrence of highly pathogenic avian influenza virus H5N1 in domestic poultry in Asia: a spatial multicriteria decision analysis approach. Spatial and spatio-temporal epidemiology. 2013; 4:1–14. https://doi.org/10.1016/j.sste. 2012.11.002 PMID: 23481249
- 94. Refsgaard JC, van der Sluijs JP, Brown J, van der Keur P. A framework for dealing with uncertainty due to model structure error. Advances Water Resources. 2006; 29:1586–97.
- Refsgaard JC, van der Sluijs JP, Hojberg AL, Vanrolleghem PA. Uncertainty in the environmental modelling process—a framework and guidance. Environmental Modelling Software. 2007; 22:1543– 56.
- Franklin J. Model evaluation. Mapping Species Distributions: Spatial Inference and Prediction. Cambridge, New York: Cambridge University Press; 2009.
- 97. Muller G, Schlein Y. Nectar and honeydew feeding of Phlebotomus papatasi in a focus of Leishmania major in Neot Hakikar oasis. J Vector Ecol. 2004; 29(1):154–8. Epub 2004/07/23. PMID: 15266752.
- Boussaa S, Guernaoui S, Pesson B, Boumezzough A. Seasonal fluctuations of phlebotomine sand fly populations (Diptera: Psychodidae) in the urban area of Marrakech, Morocco. Acta Trop. 2005; 95 (2):86–91. Epub 2005/06/30. https://doi.org/10.1016/j.actatropica.2005.05.002 PMID: 15985259.
- 99. Coleman RE, Burkett DA, Sherwood V, Caci J, Spradling S, Jennings BT, et al. Impact of phlebotomine sand flies on U.S. Military operations at Tallil Air Base, Iraq: 2. Temporal and geographic distribution of sand flies. J Med Entomol. 2007; 44(1):29–41. <u>https://doi.org/10.1603/0022-2585(2007)44[29:</u> iopsfo]2.0.co;2 PMID: 17294918
- Boussaa S, Pesson B, Boumezzough A. Phlebotomine sandflies (Diptera: Psychodidae) of Marrakech city, Morocco. Ann Trop Med Parasitol. 2007; 101(8):715–24. Epub 2007/11/22. <u>https://doi.org/10. 1179/136485907X241398 PMID: 18028733.</u>
- Didan K. MOD13C2 MODIS/Terra Vegetation Indices Monthly L3 Global 0.05Deg CMG V006. In: Center NELPDLPDAACUSGSUEROaSE, editor.: Integrated Climate Data Center, University of Hamburg; 2015.
- 102. Pesaresi MF, Sergio. GHS Settlement grid following the REGIO model 2014 in application to GHSL Landsat and CIESIN GPW v4-multitemporal (1975-1990-2000-2015). In: European Commission JRCJ, editor. 2016.
- 103. Hazratian T, Rassi Y, Oshaghi MA, Yaghoobi-Ershadi MR, Fallah E, Shirzadi MR, et al. Phenology and population dynamics of sand flies in a new focus of visceral leishmaniasis in Eastern Azarbaijan Province, North western of Iran. Asian Pac J Trop Med. 2011; 4(8):604–9. <u>https://doi.org/10.1016/</u> S1995-7645(11)60156-9 PMID: 21914536.
- 104. Kassem HA, El-Sayed YA, Baz MM, Kenawy MA, El Sawaf BM. Climatic factors influencing the abundance of Phlebotomus papatasi (Scopoli) (Diptera: Psychodidae) in the Nile Delta. J Egypt Soc Parasitol. 2009; 39(1):305–16. PMID: 19530629.
- 105. Reza FM, Mansour N. Entomological studies of Phlebotomus papatasi and P. sergenti (Diptera: Psychodidae) as vectors of cutaneous leishmaniasis in Shiraz, Iran. Southeast Asian J Trop Med Public Health. 2006; 37 Suppl 3:115–7. Epub 2007/06/06. PMID: 17547064.
- 106. Weier J, and Herring D. Measuring Vegetation (NDVI & EVI) 2000 [14/07/2017]. Available from: https://earthobservatory.nasa.gov/Features/MeasuringVegetation/.
- 107. Doha S, Shehata M, El Said S, El Sawaf B. Dispersal of Phlebotomus papatasi (Scopoli) and P. langeroni Nitzulescu in El Hammam, Matrouh Governorate, Egypt. Annales de parasitologie humaine et comparée. 1991; 66(2):69–76.
- **108.** Quate LW. Phlebotomus sandflies of the Paloich area in the Sudan (Diptera, Psychodidae). J Med Entomol. 1964; 1(3):213–68.

© 2020 Takahashi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License:

http://creativecommons.org/licenses/by/4.0/(the "License"), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. Notwithstanding the ProQuest Terms and Conditions, you may use this content in accordance with the terms of the License.

